Author's response to reviews

Title: Anti-N-methyl-D-aspartate Receptor Encephalitis Associated with an Ovarian Teratoma: Two Cases Report and Anesthesia Considerations

Authors:

Haiyang Liu (oceanliu1979@sina.com)
Minyu Jian (yum1988@163.com)
Fa Liang (liangfa1001@yahoo.com)
Hongli Yue (yueya_1974@sohu.com)
Ruquan Han (hanrq666@aliyun.com)

Version: 8 Date: 30 September 2015

Author's response to reviews: see over
Dear Editors,

Re: Manuscript reference No. 9762394931713864

Please find attached a revised version of our manuscript “Anti-N-methyl-D-aspartate Receptor Encephalitis Associated with an Ovarian Teratoma: Two Cases Report and Anesthesia Considerations”, which we would like to resubmit for publication as a case report in BMC Anesthesiology.

Your requirements were highly insightful and enabled us to greatly improve the quality of our manuscript. The following are our point-by-point responses to each of your requirements.

We hope that the revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript be accepted for publication in BMC Anesthesiology.

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Ruquan Han, M.D., Ph.D., Professor and Chairman
Address: Department of Anesthesiology, Beijing Tiantan Hospital, Capital Medical University.
No. 6 Tiantan Xili, Beijing 100050, China.
Ph: 8610-67096660
E-mail: hanrq666@aliyun.com

Responses to Editorial requirements:

-Please format your manuscript accordingly, as it currently has a Discussion section.

According to Instructions for Authors, “Case presentation should present all details concerning the case, as well as a discussion with references to the literature, and this section may be broken into subsections with appropriate subheadings”. Further more, we referenced the newest published case reports in BMC Anesthesiology: (1) BMC Anesthesiology 2015, 15:90 (10 June 2015), (2) BMC Anesthesiology 2015, 15:108 (28 July 2015). In our understanding of above, we have moved the discussion to Case presentation section as a subsection (begin in line 146, page 7). However, English is
not our native language, it is hard for us to make sure whether such formatting is satisfactorily addressed or not. Please don't hesitate to contact me if my manuscript have any problems regarding your requirements.

- Please confirm that consent was obtained from both patients for publication of the case reports.

Yes, we confirmed the consent was obtained from both patients for publication of the case reports. And we changed our consent statement to: "Written informed consent was obtained from the patients for publication of this Case report and any accompanying images. A copy of the written consents is available for review by the Editor of this journal. (in line 243-246, page 13)" The sentences about additional files have been removed.

- Please remove additional files 1 and 2.

Additional files 1 and 2 have been removed.

- Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals).

Yes, this point is important. We have revised our manuscript as journal style.